

Author's response to reviews

Title: Hospital-acquired Fever in Oriental Medical Hospitals

Authors:

Soo-youn Moon (sooyounmoon78@gmail.com)

Ki-Ho Park (parkkiho@hotmail.com)

Mi Suk Lee (mslee@khmc.or.kr)

Jun Seong Son (isonjs@naver.com)

Version: 3 Date: 29 Jan 2018

Author's response to reviews:

1. Please note that all manuscripts must contain all the following sections under the heading 'Declarations'. The Declarations should follow the Conclusions section, and be before the References.

Abbreviations

Ethics approval and consent to participate

Consent for publication

Availability of data and material

Competing interests

Funding

Authors' contributions

Acknowledgements

→ I've added 'Declarations' after Discussion.

2. Please include the email addresses of all authors in the main text of your manuscript

→ I've added the e-mail addresses of all authors at the Title page.

3. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables should also be clean versions.

→ I've checked that it is single, clean version.